Author’s response to reviews

Title: An Intervention Program to Reduce the Number of Hospitalizations of Elderly Patients in a Primary Care Clinic

Authors:

Roni Peleg (pelegr@bgumail.bgu.ac.il)
Yan Press (yanp@zahav.net.il)
Maya Asher (mayas@clalit.org.il)
Tatiana Pugachev (pelegr@bgu.ac.il)
Hadas Glicensztain (pelegr@clalit.org.il)
Mila Lederman (ronipel@gmail.com)
Aya Biderman (sbider@netvision.net.il)

Version: 4 Date: 21 December 2007

Author’s response to reviews:

To the editor,

We would like to thank you for the helpful comments and suggestions for revision # 2.

In accordance with your suggestion, and that of Reviewer #1, we have done the following revisions (highlighted in yellow in the manuscript):

1. As suggested by you, we added a brief description of the intervention to the Abstract in the Methods section.

2. As requested by Reviewer #1 we added to the Discussion section a clarification of the justification of the approach to organizational change and added 3 new references (#14-16 in the new version-R2).

We hope that you find our manuscript, in its revised version (R2) worthy of publication in your journal.

Sincerely,

Roni Peleg, MD